

GOPEN ACCESS

Citation: Brady OJ, Kucharski AJ, Funk S, Jafari Y, Loock MV, Herrera-Taracena G, et al. (2021) Casearea targeted interventions (CATI) for reactive dengue control: Modelling effectiveness of vector control and prophylactic drugs in Singapore. PLoS Negl Trop Dis 15(8): e0009562. https://doi.org/ 10.1371/journal.pntd.0009562

Editor: Hannah E. Clapham, University of Oxford, UNITED KINGDOM

Received: December 16, 2020

Accepted: June 14, 2021

Published: August 11, 2021

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0009562

Copyright: © 2021 Brady et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data are available from [25], and as part of the DENSpatial R package

RESEARCH ARTICLE

Case-area targeted interventions (CATI) for reactive dengue control: Modelling effectiveness of vector control and prophylactic drugs in Singapore

Oliver J. Brady^{1,2}*, Adam J. Kucharski^{1,2}, Sebastian Funk^{1,2}, Yalda Jafari^{1,2}, Marnix Van Loock³, Guillermo Herrera-Taracena⁴, Joris Menten⁵, W. John Edmunds^{1,2}, Shuzhen Sim⁶, Lee-Ching Ng⁶, Stéphane Hué^{1,2‡}, Martin L. Hibberd^{7‡}

1 Centre for Mathematical Modelling of Infectious Diseases, London School of Hygiene & Tropical Medicine, London, United Kingdom, 2 Department of Infectious Disease Epidemiology, Faculty of Epidemiology and Public Health, London School of Hygiene & Tropical Medicine, London, United Kingdom, 3 Janssen Global Public Health, Janssen Pharmaceutica NV, Beerse, Belgium, 4 Janssen Global Public Health, Janssen Research & Development, LLC, Horsham, Pennsylvania, United States of America, 5 Quantitative Sciences, Janssen Pharmaceutica NV, Beerse, Belgium, 6 Environmental Health Institute, National Environment Agency, Singapore, Singapore, 7 Department of Infection Biology, Faculty of Infectious Tropical Diseases, London School of Hygiene & Tropical Medicine, London, United Kingdom

Both authors jointly supervised the work.
* oliver.brady@lshtm.ac.uk

Abstract

Background

Targeting interventions to areas that have recently experienced cases of disease is one strategy to contain outbreaks of infectious disease. Such case-area targeted interventions (CATI) have become an increasingly popular approach for dengue control but there is little evidence to suggest how precisely targeted or how recent cases need to be, to mount an effective response. The growing interest in the development of prophylactic and therapeutic drugs for dengue has also given new relevance for CATI strategies to interrupt transmission or deliver early treatment.

Methods/Principal findings

Here we develop a patch-based mathematical model of spatial dengue spread and fit it to spatiotemporal datasets from Singapore. Simulations from this model suggest CATI strategies could be effective, particularly if used in lower density areas. To maximise effectiveness, increasing the size of the radius around an index case should be prioritised even if it results in delays in the intervention being applied. This is partially because large intervention radii ensure individuals receive multiple and regular rounds of drug dosing or vector control, and thus boost overall coverage. Given equivalent efficacy, CATIs using prophylactic drugs are predicted to be more effective than adult mosquito-killing vector control methods and may even offer the possibility of interrupting individual chains of transmission if rapidly deployed. CATI strategies quickly lose their effectiveness if baseline transmission increases or case detection rates fall.

(https://github.com/obrady/SpatialDengue/ releases/tag/V1.2).

Funding: This work was funded by a grant from Janssen (grant number ITPMZG4810) to MH. OJB [206471/Z/17/Z], AJK [206250/Z/17/Z] and SF [210758/Z/18/Z] are sponsored by the Wellcome Trust (<u>https://wellcome.ac.uk/</u>). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: MVL, GHT and JM are employees of Johnson & Johnson. LS and LN are employees of the Singapore national Environment Agency

Conclusions/Significance

These results suggest CATI strategies can play an important role in dengue control but are likely to be most relevant for low transmission areas where high coverage of other non-reactive interventions already exists. Controlled field trials are needed to assess the field efficacy and practical constraints of large operational CATI strategies.

Author summary

In resource limited settings there is a pressing need for more efficient, more targeted ways of controlling transmission and preventing outbreaks. One option is to use case-area targeted interventions (CATI) that are focussed on areas that have recently reported disease cases. The effectiveness of such CATI strategies is highly dependent on how the disease spreads. Despite CATI strategies being widely used to control the vector-transmitted disease dengue, little evidence underpins its effectiveness.

In this analysis we formulate a mathematical model designed to test the effectiveness of CATI strategies for dengue control in Singapore- a best case test scenario for the approach. Simulation from this model suggested CATI are likely to be effective for dengue, but need to have large (250m+) radii around index cases and may not be suitable in higher transmission areas.

These results, when combined with limited field evidence of efficacy, suggest that CATI strategies are unlikely to be universally applicable dengue control tools. Only once high coverage with other (non-reactive) interventions has been achieved and comprehensive rapid disease surveillance has been established are CATI strategies likely to become efficient methods of disease control.

Introduction

Dengue is a viral disease vectored by the *Aedes aegypti* mosquito that causes an estimated 100–400 million infections per year worldwide [1, 2]. While most dengue virus (DENV) infections only result in mild disease, the ubiquity of infection causes significant morbidity and frequently overwhelms healthcare facilities. The arboviruses of chikungunya and Zika which are also transmitted by the *Aedes aegypti* mosquito threaten to establish permanent transmission cycles if current dengue outbreaks cannot be rapidly and effectively controlled.

No country has been able to permanently eliminate DENV transmission and focus has shifted towards control and, in particular, preventing big outbreaks as early as possible [3]. One strategy recommended to achieve this is the use of case-area targeted interventions (CATI), where interventions are targeted in and around (typically within 100m) the houses of dengue patients after they present to healthcare facilities [3]. For dengue, these interventions are currently restricted to vector control activities including fogging to kill adult mosquitoes and source reduction to eliminate mosquito egg-laying habitat [4]. CATI strategies are becoming increasingly popular given the need to optimise the use of limited vector control resources and the challenges of sustaining area-wide control [5, 6]. However, the evidence-base for the speed, scale and efficacy of response required for CATI strategies to contain a dengue outbreak remains far from complete. The suggestion that the control radius should equal 100m is largely based on maximum *Ae. aegypti* mosquito flight range estimates from mark-release-recapture studies, which may not cover sufficiently large areas to properly document maximum dispersal [6]. Such studies may also not be representative, and longer flight ranges (>300m) have been documented [7]. In addition, there is strong evidence that urban spread of dengue is largely

attributable to movement of infected humans [5, 8, 9] and response time recommendations are based more on practical realities than quantitative estimates of likely effectiveness [3].

While no drug is currently used for dengue treatment or prevention, a range of short-acting anti-viral compounds are in development that could protect against infection, reduce human infectiousness and decrease the severity of disease [10]. Such prophylactic drugs could play a key role in CATI strategies if their deployment can be optimised and appropriately combined with vector control.

As dengue surveillance systems have become increasingly electronic and automated, there is renewed interest in using real-time CATI strategies for dengue control. However, three features of dengue epidemiology may limit the effectiveness. First, a high proportion of DENV infections are not detected by routine surveillance systems [11], meaning foci of infection may be missed. Second, index case cluster studies have shown that humans peak in infectiousness at the time of, or just before, becoming symptomatic [12] meaning the response may be too slow to prevent onwards transmission. Third, *Ae. aegypti* mosquitoes usually bite during the day when people are more mobile, meaning the place of residence may not accurately predict the location where they were infected.

No randomized control trials have been undertaken for CATI against adult Ae. aegypti [6], but there is evidence to suggest that dengue infection is highly spatiotemporally clustered either in the household or within the local area (up to 1km) [5, 13-15]. In the absence of locally applicable field trials, mathematical models can be used to predict what effect response speed and scale will have on the effectiveness of a CATI strategy and identify what role novel prophylactic drugs could play in improving outbreak control. Two theoretical modelling studies of spatially reactively targeted interventions (CATI) for outbreak control have been undertaken for dengue [16, 17], but have only considered simplistic descriptions of human movements between two patches, i.e. travel between work or home or between village and city. One more detailed modelling study for chikungunya [18] suggests that the effectiveness of CATI is likely to be highly sensitive to the proportion of symptomatic cases, which is known to be lower for dengue [1, 19, 20]. Finally, one arbovirus transmission model has been recently used to compare index case-targeted vector control with realistic human movements. This found that the performance of CATI strategies is highly dependent on the human density of the area concerned, but this analysis was limited to more spatially coarse neighbourhood-level analyses [21]. None of these analyses have been formally validated in their ability to reproduce the unique spatial clustering patterns of urban dengue outbreaks and no modelling analyses have considered the potential use of prophylactic drugs for targeted outbreak control.

Here we test the potential effectiveness of vector control and prophylactic drug-based CATI strategies for the control of dengue outbreaks and identify the operational speed and scale of response required. For this, we use a spatially explicit dengue dataset describing a series of outbreaks in Singapore between 2013–2016. Singapore has a robust vector control programme that maintains low baseline transmission intensity, a strong case surveillance system that detects a relatively high proportion of DENV infections in a timely manner and has readily available geopositioned dengue case data over multiple years to characterise transmission [22–24]. These characteristics make such a location a best-case scenario for CATI strategies for dengue control.

Methods

Data

We used weekly data on the location of cases across Singapore for the time period May 2013 – June 2016 as reported by the Singapore National Environment Agency (https://www.nea.gov. sg/dengue-zika/dengue/dengue-clusters, S1 Fig). A total of 82,185 geopositioned suspected

and confirmed cases occurred over this time period with peaks in early-to-mid 2013, 2014 and 2016. The data are available from [25], and as part of the DENSpatial R package (https://github.com/obrady/SpatialDengue/releases/tag/V1.2).

To represent seasonal fluctuations in transmission intensity from the mosquito population, data on adult mosquito abundance was obtained from the National Environment Agency entomological surveillance programme over the same time period and is included as part of the DENSpatial R package. The average weekly Gravitrap *Aedes aegypti* index (GAI–number of female aegypti caught per functional trap) across 34 representative sentinel sites measures seasonal abundance across Singapore (SI1.1) [26].

To generate high resolution population maps building-level data from OpenStreet (www. openstreetmap.org) was combined with planning area-level population estimates from the Singapore department of Statistics 2020 population trends report [27]. Areas where humans do not spend significant portions of time were excluded then information of build size and height were extracted to use as a basis for apportioning planning-area level population counts (S1 Text). This population raster was then aggregated to 1km x 1km pixel level then rounded to the nearest integer for all subsequent model-based analyses. This 1km x 1km patch size choice was based on previous work that suggests dengue transmission in large outbreaks tends to be highly spatially clustered up to 1km) [5, 13–15]. For clarity, this does while the model is implemented on a 1km x 1km patch scale, the model is still able to track finer-scale transmission events and intervention policies by assuming individuals homogeneous mix and that features of the environment (mosquito population size and human immunity) are broadly similar at sub 1km x 1km scales. We believe to be an acceptable approximation for assessing the effects of CATI interventions during large outbreaks.

Model structure

The model used for this analysis is a spatially explicit patch based stochastic model. Within each patch, humans are modelled in Susceptible, Infectious and Recovered compartments (SIR) with infectiousness varying over time since becoming infected (Fig 1). We assume humans peak in infectiousness at the timing of onset of symptoms (i.e., after completion of the intrinsic incubation period (IIP)). Following the IIP they develop either apparent illness and are detectable or remain inapparent until 8 days post symptom onset upon which they transition to the Recovered and immune compartment. Humans can also move to a temporarily immune compartment (Rp) if treated with prophylactic drugs. Mosquitoes also have an additional exposed but not infectious compartment but with no recovery from infection (SEI). To capture realistic time delays in transmission and detection, intrinsic (human) and extrinsic (mosquito) incubation periods are included and infected humans can only be detected as symptomatic cases after completing their IIP. We assume that all humans contribute to transmission the same regardless of their symptoms or whether they were detected as cases.

Seasonal variation in transmission intensity is represented in our model using data from weekly adult mosquito collections (GAI, see data section). Temporal variations in GAI over the relevant time period were scaled to have a mean equal to one and a floor of 0.1 (assuming transmission is possible even at the most unsuitable times of year) then was used to multiply the transmission coefficient for the corresponding day of the year. We assume that outbreaks occur in this setting due to a combination of seasonal effects and more permanent (up to 1 year) changes to local environments that increase human-mosquito contact rates, for example, construction site development [22].

While transmission dynamics are modelled at the patch level, human populations are modelled as continuously mobile, allocating their time to a number of patches including their home



Fig 1. Mathematical model structure for each patch. Humans (h) and mosquitoes (m) are divided into susceptible (S), exposed (Mosquito-only, E), Infectious (I) and recovered and immune (R) compartments. Humans can become temporarily immune (Rp) if treated with prophylactic drugs with effective coverage rate r_d with protection waning at rate c_d . Infection is controlled by a mosquito-human contact rate (β) after which humans and mosquitoes go through an incubation period (at rates ε_h and ε_m respectively). Humans then naturally recover after $1/\gamma$ days of illness while mosquitoes stay infectious for life. Mosquitoes die at a natural death rate μ_n but can also be subject to an additional focal control mortality rate μ_c . All human population compartments are transient and made up of all individuals who spend time in the patch. Mosquitoes do not move between patches.

residence. Here we represent flow of humans between patches using three different movement models: i) exponential, ii) gravity and iii) radiation to represent simple distance-based, sourcesink and structured commuting-type human movement behaviours respectively [28, 29] (equations in <u>S2 Text</u>). The choice of human movement model is estimated during the model fitting procedure. We assume mosquitoes do not move between patches (patch size 1km x 1km) consistent with the limited dispersal of the primary species vector in Singapore (*Ae. aegypti*) [22].

Despite the co-circulation of multiple serotypes in Singapore both past and present, the outbreaks between 2013 and 2015 were predominantly due to DENV1 (70–90% of reported cases), as were the majority of cases leading up to the period of interest with the exception of a brief period of DENV2 circulation 2007–2012 [30, 31]. With a focus on only modelling the dynamics of the outbreaks in question we assume only a single serotype model with sterilising infection. Given the existence of temporary cross serotype immunity following recent infection with any one dengue type and the predominance of one type, we would expect similar results from a multi-serotype model over the short timeframe over which our model is focussed.

Transmission dynamics

Mosquitoes within each patch (*i*) exist in one of three states: Susceptible (S^m), Infected (but not yet infectious) (E^m) and Infectious (I^m). Transitions between these states are determined by the following equations:

$$\begin{split} dS^{m}/dt &= (\mu_{n} + \mu_{c})(S^{m} + E^{m} + I^{m}) - \omega S^{m} - \mu_{n}S^{m} - \mu_{c}S^{m} \\ dE^{m}/dt &= \omega S^{m} - \mu_{n}E^{m} - \mu_{c}E^{m} - \varepsilon_{m}E^{m} \\ dI^{m}/dt &= \varepsilon_{m}E^{m} - \mu_{n}I^{m} - \mu_{c}I^{m}, \end{split}$$

Where μ_n and μ_c are the rates mosquito mortality due to natural and extra CATI measures respectively (i.e., we assume vector control measures reduce the number of adult mosquitoes), ε_m is the rate of EIP completion and ω is the risk of infection for each susceptible mosquito in the patch. Calculating ω_i involves summing the total person hours spent in patch *i* by both human residents $(T_{i\rightarrow i})$ and visitors $(T_{j\rightarrow i})$. Infectivity of any given human is a function of the mosquito-human contact rate of patch *i* on day z ($\beta_{i,z}$) and the individual infectiousness of each human which is dependent on the number of days since becoming infected (θ_d).

$$\omega_{i} = \sum_{j} \left[T_{j \to i} \left(\frac{\beta_{i,z} \theta_{d} I_{j,d}^{h}}{N_{j}^{h}} \right) \right]$$

Consistent with observations of viral titre in dengue patients [12] over time, we assume normally distributed infectiousness, peaking at symptom onset (where transmission probability = 1), standard deviation of 2 and constrained to 0 at time of infection and eight days post symptom onset. The timing of peak viremia is determined by a lognormal IIP (ε_h) [32].

$$\theta_d = N(\boldsymbol{\varepsilon}_h, 2)$$

$$\boldsymbol{\varepsilon}_{h} = Lognormal(\mu, \sigma^{2})$$

The additional risk of mortality due to CATI mosquito control efforts (μ_c) is applied if the patch (*i*) is within the defined radius (*L*) of an index case (*j*).

$$egin{aligned} &l_{min} = \min(l_{i
ightarrow 1}, \ldots, l_{i
ightarrow j}) \ &f(\mu_c) = \left\{ egin{aligned} &\mu_c ext{ if } l_{min} \leq L \ &0 ext{ if } l_{min} > L \end{aligned}
ight. \end{aligned}$$

Infection dynamics in humans were modelled in Susceptible (S^h), Infectious (I^h), recovered due to natural infection (R^h) and temporarily immune due to prophylactic drugs (R_p^h). Transitions between these states are as follows:

$$\begin{split} dS^{h}/dt &= rR_{p}^{h} - \varphi_{i}S^{h} - c_{d}S^{h} \\ \frac{dI^{h}}{dt} &= \varphi_{i}S^{h} - c_{d}I^{h} - I_{d=\frac{1}{e_{h}+\frac{1}{\gamma}}}^{h} \\ dR^{h}/dt &= I_{d=\frac{1}{e_{h}+\frac{1}{\gamma}}}^{h} + c_{d}I^{h} \end{split}$$

$$dR_p^h/dt = c_d S^h - rR_p^h$$

 $S^h + I^h + R^h + R_p^h = N^h$

Susceptible humans can either be infected in their resident (*i*) patch or any of the patches they visit (*j*):

$$\varphi_{i} = \sum_{j} \left[T_{i \rightarrow j} \frac{\beta_{j,z} I_{j}^{m}}{\sum_{k} (T_{k \rightarrow j} N_{k}^{h})} \right]$$

All individuals (*i*) who reside within a distance *L* of the home location of any detected dengue case (*j*) will receive prophylactic drugs with an effective coverage level c_d .

$$l_{min} = \min(l_{i \to 1}, \dots, l_{i \to j})$$
$$f(c_d) = \begin{cases} c_d \text{ if } l_{min} \leq L\\ 0 \text{ if } l_{min} > L \end{cases}$$

Effective coverage includes barriers to access, adherence, eligibility and efficacy of the prophylactic drug in question. Individuals who are treated with drugs when already infected are assumed to acquire sterilizing immunity akin to natural infection. Infected individuals remain in the Infectious state for the duration of their IIP + 8 days of disease $(1/\gamma)$ before transitioning to recovered state with sterilizing immunity. Individuals can be detected (D_t^h) at any point in their symptomatic infectious stage (i.e., $1/\varepsilon_h < d < (1/\varepsilon_h + 1/\gamma)$]) with daily detection probability δ which is inferred from the case data (N.B. overall probability that a case is detected is therefore 8δ):

$$D^h_t = \delta I^h_{rac{1}{ch} < d < \left[rac{1}{ch} + 8
ight]}$$
 $m{arepsilon}_{h,d} = Lognormal(\mu,\sigma^2)$

The effective reproductive number for each patch was equal to the product of the total number of bites an infectious person receives over their duration of their infectious period (θ), the probability each infected mosquito survives beyond the virus' EIP [33] and the number of infectious bites delivered to susceptible humans by the infected mosquito population post EIP:

$$egin{aligned} R_{e\!f\!f} &= rac{\hat{eta}S^m heta}{N^h} \cdot rac{m{arepsilon}_m}{\mu_n + m{arepsilon}_m} \cdot rac{\hat{eta}\overline{S_h}}{\mu_n\overline{N_h}} \ &\overline{N^h} &= \sum_j (T_{j
ightarrow i}N_j^h) \ &\overline{S^h} &= \sum_j (T_{j
ightarrow i}S_j^h) \end{aligned}$$

The model is implemented using a daily time step with events modelled as realisations from stochastic binomial processes. Where multiple additions or subtractions from state

compartments existed (e.g., Exposed mosquitoes completing EIP or dying), the order of processes were randomised then sequentially carried out to ensure each mosquito or human had only one outcome and that certain processes were not more likely than others.

Parameters and priors

Parameter estimates and priors are given in Table 1. We use fixed distributions for parameters describing DENV IIP and EIP (ε_h and ε_m) from a systematic review of transmission experiments [32]. Our estimate of the maximum duration of symptomatic illness ($1/\gamma$) corresponds to the upper limit of estimates of the length of the febrile phase of illness [3]. For other parameters we assign weekly informative priors that are either uniform (k, δ, β) or broadly distributed (μ_n) [34]. The vector-human transmission rate (β) was split into two components; a mean (β_μ) and a degree of spatial correlation with the existing levels of immunity in the population (β_p). Because our fitting metrics focussed on the relative distribution of cases over space and time, the daily probability of detection of an infectious human (δ) was used as a scaling parameter to match reported case count post-hoc after other parameters had been fitted. No constraints were put on the daily probability of detection other than $0 \le \delta \le 1$.

In the absence of an internationally recognised target product profile for dengue prophylactics, here we consider a drug with 90% protective efficacy against infection administered with 90% coverage to all individuals regardless of infection status ($c_d = 0.81$). We assume that the drug also acts as a therapeutic, preventing transmission and symptoms if given to someone already infected with DENV. We assume the drug is taken with sufficient frequency to ensure

Table 1. Model parameter values.

Parameter	Definition	Value	Source					
Fixed parameters and constraints								
$1/\varepsilon_h$	Intrinsic incubation period of the virus	$LogNormal(\mu = 5.9, \sigma = 1.05)$	[32]					
$1/\varepsilon_m$	Extrinsic incubation period of the virus	$LogNormal(\mu = 7, \sigma = 1.23)$	[32]					
θ	Cumulative human infectiousness over the course of an infection	$\theta = \int \frac{negBin(29.37,0.88)}{\max(negBin(29.37,0.88))}$	[<u>35</u>], <u>S2</u> Fig					
$1/\gamma$	Maximum duration of symptomatic illness following completion of IIP	8 days	[3]					
$R_{t,0}$	Average effective reproduction number at the beginning of the outbreak	$1 < R_{eff,t = 0} < 10$	[36]					
1/ <i>r</i> _d	Duration of effectiveness of prophylactic drugs and the number of days for which vector control is effective	Assumed 30	-					
Fitted parameters with priors								
k	Proportion of time at risk of dengue infection ^{**} each human spends at home in patch <i>i</i> vs all other locations	$\mathcal{U}(0,1)$	-					
δ	Daily probability of a DENV infected individual being detected and reported as a case	$0 \le \delta \le 1$	-					
μ_n	Daily mortality rate of a mosquito	Beta($\alpha = 11.93, \beta = 107.4$)	[34]					
β	Vector-to-human and human-to-vector transmission rate*		-					
τ	Type of human movement model	"gravity", "radiation" or "exponential" (<u>S2</u> Text)	-					
Experiment	ally varied parameters	·						
Cd	The effective coverage of prophylactic drugs in the target area	0.81 (0.2–0.81)	-					
μ _c	The effective coverage of adult mosquito vector control in the target area	0.81 (0.2–0.81)	-					
L	Radius around an index case that receives drug dosing or vector control	50-1000 meters	-					

* Vector-human transmission rate and its variation over space is parameterised using mean (μ) and correlation with baseline immunity (p) parameters.

** The total time at risk of dengue infection can equal less than 24 hours e.g., only at risk at dawn and dusk, this parameter merely divides however many hours are at risk among different patches.

https://doi.org/10.1371/journal.pntd.0009562.t001

30 days of effective prophylaxis with complete adherence. For comparability, we assume that each application of vector control is effective for 30 days, has a 90% probability of killing adult mosquitoes on each day and is applied with 90% coverage across the target area. For index case radii < 1000m we assume a proportional reduction in coverage that lowers effective coverage.

Initial conditions

A nationally-representative age-stratified serosurvey for dengue was conducted between December 2013 and February 2014 just before the two main focus time periods of our data clusters [37]. In this study Tan et al. found a nationwide average IgG prevalence of 48.6%, with clear increases with age. Despite the authors finding no geographic variation in age-specific IgG prevalence rates, because the average age of residents varies across Singapore, the observed IgG prevalence in the resident population will still vary. To represent this geographic variation we combined the age-specific IgG seroprevalence measurements from Tan et al. with age-stratified population data from Singapore Department of Statistics 2020 population trends report [27] (S3 Fig). Dengue virus serotype 1 has predominated (accounting for 70–90% clinical cases) since Singapore began DENV serotype surveillance with the exception of the years of 2007–2012 [24, 31]. We therefore consider an average IgG seroprevalence of 48.6% representative of the degree of functional immunity to dengue in Singapore in 2014 despite the co-circulation of other serotypes at lower prevalence during this time.

The susceptible mosquito population size was initialized with an inexhaustible 1:1 ratio with human population size. Model fitting was insensitive to initial mosquito population size as the magnitude and spatial variation in mosquito-human contact rate can compensate for any true differences in population size. The number of infected humans in the previous transmission generation was calculated from the first timepoint of the fitting dataset with the total number of infectious and exposed mosquitoes derived using the following equations:

$$egin{aligned} &I_{0-g}^h = D_0^h/8\delta R_t \ &E_0^m = I_{0-g}^h hetaeta_z \ &I_0^m = E_0^m(1-\mu_n)^{\left(rac{1}{arepsilon_m}
ight)} \end{aligned}$$

Infectious and exposed mosquitoes were randomly distributed in space weighted by the product of each infectious persons' time allocation between patches and variation in infectiousness given days since becoming infected. Because cases were only reported weekly, we randomly assigned a day to each reported case over the week prior to their notification.

Model fitting

Because the time series data we had access to began mid-epidemic and because Singapore often sees frequent importation and co-circulation of multiple chains of transmission [38], we pre-processed the data using a data clustering algorithm prior to model fitting. This clustering algorithm grouped reported cases into transmission clusters based on their timing and location. For this analysis a simple two week generation time was assumed [39] and data from three weeks ago was used when data from two weeks prior was missing. Parent cases were identified by finding the nearest case from two weeks ago with distance measured using i) exponential, ii) gravity or iii) radiation model networks (generating three datasets). Cases were considered to belong to a new case cluster (following importation) if the distance of the nearest

case was above the 99th centile of case distance measurements. A sensitivity analysis to this choice of linkage threshold was performed at 95th and 99.9th centiles (S1 Table). The three most probable clusters were selected for model fitting (3 clusters x 3 movement models = 9 fitting datasets). To focus on the dominant period of transmission, each cluster was restricted to either the 2014 (Weeks 40–99) or 2015 (Weeks 100–160) dengue season based on height of the peak.

The model was fitted to each movement model cluster (x_τ) combination using sequential Monte Carlo approximate Bayesian computation (SMC ABC) following an algorithm from Toni et al. [40]. We fit the model by minimising mean squared error of the relative distribution of cases across eight equally spaced time periods throughout the epidemic (*T*) and eight quadrants of Singapore (*Q*):

$$egin{aligned} & d_{x_{ au}} = T^{rel} + S^{rel} \ & T^{rel} = \sum_{T=1}^{T=8} \left(D_T^{h,observed} - D_T^{h,predicted}
ight)^2 \ & S^{rel} = \sum_{Q=1}^{Q=8} \left(D_Q^{h,observed} - D_Q^{h,predicted}
ight)^2 \end{aligned}$$

Three rounds of SMC were performed with each round consisting of 1,000 parameter samples, each evaluated from the median prediction from 10 model simulations. The top 10% parameter sample combinations (lowest *d* values) were retained and used for the proposal distribution in the next SMC round. Only combinations of samples that gave an effective reproductive number between 1 and 10 were accepted to increase computation efficiency. For round one, parameter samples were taken from the priors in Table 1. For rounds two and three parameter combinations from the previous round were sampled then individually uniformly perturbed up to $\pm 10\%$.

Results

There is a limited time window for preventing individual transmission events, but a prophylactic drug could extend this

To determine the effective time window for CATI strategies we first constructed a simple model of DENV generation time in one human host and their vectors (Fig 2). This simple model assumed that vector control and prophylactic drugs had 100% efficacy and that delays in their application was the only factor that limited effectiveness. A limiting factor for reactive dengue control is that humans only become symptomatic at the time of, or just before, their peak in viraemia. As dengue is typically a mild illness, especially in the early stages, symptomatic individuals often take several days to seek treatment, receive a correct diagnosis and be notified to the public health authorities (detection delays). Further delays in notification of the case, its details and the organisation of a house/area visit (response delays) can also mean reactive control can be implemented days or even weeks after the transmission event.

We find that even short delays in detection or response can have a significant impact on the probability of controlling onward transmission. The probability of preventing onward transmission by killing adult mosquitoes (with, for example, fogging) rapidly declines with time; a response nine days after symptom onset is approximately half as effective as a response on the day of detection (Fig 2B). By contrast, adult vector control responses deployed greater than



Fig 2. Probability of interrupting a single chain of transmission over time. A) the probability distribution of infectious humans, the mosquitoes they infect and the subsequent human dengue cases. The red dotted line shows a typical reactive delay. Day 0 is the day of first symptoms due to dengue. B) Probability of interrupting transmission over different delay durations. Black dotted lines show the time when there is a 50% chance of interrupting transmission.

two weeks after the patient's first symptoms have less than a 20% chance of interrupting onwards transmission.

CATI strategies with prophylactic drugs have the potential to extend this tight time window of effective control by as much as the human IIP (~ 6 days), depending on how such a prophylactic drug prevented disease and transmission. Averting disease cases with reactive control using prophylactic drugs could be effective (> 50% chance of interrupting transmission chains) up to 13 days post symptom onset (Fig 2).

While this simple model gives insight into how reactive delays affect the relative effectiveness of controlling a single chain of transmission, additional factors complicate predicting absolute effectiveness of such strategies. This is because: i) it assumes that control can be accurately targeted to the area(s) of infected mosquitoes (vector control) or infected humans (drugs) which may not be the case as infection events typically happen in the day when many humans may not be at their place of residence and ii) it assumes that all infections can be detected when in reality many dengue infected individuals are asymptomatic or don't seek treatment. The degree of spatial clustering of successive generations of DENV transmission affect the importance of each of these factors. Therefore, to predict effectiveness in more detail we constructed a spatially explicit stochastic model fit to dengue outbreak data from Singapore and simulated the effectiveness of different vector control and drug CATI strategies.

Model fit results

As a data pre-processing step, we first disaggregated reported cases in Singapore from May 2013 –June 2016 into distinct clusters using a spatiotemporal case-clustering algorithm based on three different kinds of human movement models (Fig 3). There was a consensus among movement models on the emergence of three main case clusters during the period of observation. The largest (primary cluster, red Fig 3) has seen continual low-level circulation until causing a large outbreak in 2015 (peaking at week 140). During the 2014 outbreak two main clusters predominated that then fell to low levels in 2015 (Secondary and Tertiary clusters Fig 3). All cluster algorithms also identified other clusters throughout the period of observation that may represent pre-existing case clusters with an unobserved origin, importations from neighbouring countries or smaller foci of transmission. To simplify model fitting, we chose to



Fig 3. Top three identified case clusters according to exponential (A), gravity (B) and radiation (C) human movement models. Week 0 is May 2013. Only top 5 clusters shown. Dotted lines indicate the beginning of the 2014 and 2015 outbreaks.

fit to the three main clusters for each human model and truncated each cluster to the year in which it caused its major outbreak (2015 for the primary cluster and 2014 for the secondary and tertiary clusters, dotted black lines Fig 3).

A fine scale dengue microsimulation model was then fit to each of the nine cluster datasets in Fig 3 using ABC SMC. All models offered improved fit (26.7–36.8% reduction in *d*) with each SMC round giving positive but diminishing improvements (S4 Fig). The radiation human movement model gave the best fit to the first cluster (d = 2.66, Table 2) while the exponential model gave the best fit to clusters 2 and 3 (d = 2.75 and d = 2.95 respectively, Table 2).

	Exponential movement		Gravity movement		Radiation movement	
	Final deviation	% improvement	Final deviation	% improvement	Final deviation	% improvement
Cluster 1	2.90	16.64	2.69	24.04	2.66	30.11
Cluster 2	2.75	26.66	2.74	22.56	2.79	17.30
Cluster 3	2.95	36.76	3.00	34.98	3.67	11.40

Table 2. Final values of the obje	ctive function (d) across and within	clusters and its improvement between rounds	1 and 3 of sequential Monte Carlo round
		· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·

The posterior distribution of model parameters suggests differing patterns of transmission between cluster 1 (the largest cluster) and clusters 2 and 3 (Fig 4). The fitted model suggests that cluster 1 occurred among a group of less mobile individuals (mean 63% at-risk time spent at home) living in lower transmission intensity areas (mean 0.27 bites per day) but with a higher probability of reporting (0.15%).

In contrast, parameter values from cluster 2 and 3 fits suggest transmission among a more mobile set of the population (15% and 9% time at risk spent at home) in higher transmission intensity areas (0.70 and 0.68 bites per day) and a lower probability of reporting (0.04% and 0.03%). No clear correlation between transmission intensity in the current outbreak and historical transmission intensity was observed for any clusters (Transmission coefficient (correlation), Fig 4).

To assess the fitted model's ability to reconstruct the spatio-temporal patterns of transmission we compared the distribution of distances between cases in successive generations of



Fig 4. Histograms of the distribution of each model parameter using the best fitting human movement model (exponential for cluster 1, gravity for clusters 2 and 3) over successive sequential Monte Carlo rounds (Red = round 1, Green = round 2, Blue = Round 3). Probability of reporting is on a logarithmic scale (base 10).

https://doi.org/10.1371/journal.pntd.0009562.g004

transmission, i.e., dispersal kernels (Fig 5). Distance between cases two weeks apart were compared to approximate the serial interval of DENV. This analysis showed that the majority of cases in the next generation of transmission are likely to occur locally (84%, 80% and 86% for clusters 1–3) with long-distance dispersals orders of magnitude lower. The fitted models replicated comparable dispersal kernels, particularly for clusters 1 and 2. The models estimate comparable within patch transmission (77%, 70% and 61% for clusters 1–3) with a slight tendency to overestimate long distance transmission events (9000m+).

CATI strategies can be effective in both the short and long term, but effectiveness is likely to be variable, especially with vector control

To assess the effectiveness of different CATI strategies, our model was run using an ensemble of posterior parameter values from each cluster to represent the likely effectiveness under real world conditions in Singapore where multiple chains of transmission often co-circulate. The model suggests that individuals living in areas that receive at least one CATI prophylactic drug dose at any point in the outbreak show strong protection against future infections over the duration of the drug's effective period. This level of protection, on average, exceeds the effective coverage level of the drug when deployed (median 100% predicted efficacy vs 81% drug effective coverage, Fig 6A), suggesting the non-treated individuals within the area receive some indirect protection from local, rapid control. Long-term effectiveness in treated patches declines but still maintains efficacy in most areas (median 50%). Effectiveness was, however, highly variable (95CIs span 0–1) and some patches see little or no impact. Multivariate regression analysis showed that patches in low population density areas had higher effectiveness than more densely populated areas and that more connected areas treated later in the outbreak had marginally higher effectiveness (Fig 6B).

For equivalent effective coverage and duration of efficacy, CATI with vector control is predicted to have lower, less persistent and considerably more variable effectiveness (effectiveness 63% short term, 0% long term, Fig 6A). Both short and long-term effectiveness of patches treated using a CATI strategy are likely to be lower than if vector control was consistently applied in these patches throughout the outbreak, although a CATI strategy would also use fewer resources. This occurs due to reinvasion of transmission into each patch, asymptomatic infection and, to a lesser extent, the additional delay in response time of vector control interventions (Fig 2). Applying both prophylactic drugs and vector control in a CATI strategy has the potential to marginally improve long-term effectiveness (53% vs 50%).

CATI strategies with a large case radius can make up for late or low effectiveness responses

Next we tested the sensitivity of effectiveness of CATI strategies to I) the response radius around an index case, ii) the delay between case detection and intervention application and iii) the effectiveness of the intervention used. Due to the slow average spatial diffusion rate of the dengue outbreak, we observe only moderate sensitivity to delays in CATI response with only a 5–15% drop in effectiveness between a response that occurs on the day of detection compared to a response 2 weeks later (Fig 7A and 7B). Contrastingly, increasing the radius around the index case that interventions are deployed significantly increases overall effectiveness of the CATI strategy. We predict that a CATI response with a 100m radius delivered 2 weeks after detection will be more effective than a response with a 50m radius delivered on the day the case is detected, emphasising the importance of scale over speed in CATI response. CATI using vector control showed lower overall effectiveness and less sensitivity to implementation



Fig 5. Comparison of the spatial transmission kernels of observed cluster data and model predictions. These histograms show the observed and model predicted distances between detected cases in the current week and cases two weeks (approximately one serial interval) ahead. Case distances are calculated as Euclidian distance to nearest neighbour with case location assigned to the midpoint of a 1km x 1km grid. Model predictions show the median of 10 runs from the best fitting human movement model (gravity for A and B, exponential for C). Number of cases shown on a log10 scale for cluster 1 (A), 2 (B) and 3 (C).

https://doi.org/10.1371/journal.pntd.0009562.g005



Fig 6. Within patch effectiveness of case-area targeted interventions. A) Effectiveness compares proportion of symptomatic cases in areas treated with prophylactic drugs and adult vector control over the short (30 day) and long (365 day) time periods. Both drugs and vector control are predicted to have 81% effective coverage (solid horizontal line) deployed the day after index case detection within a 1km radius of the index case, maintaining efficacy for 30 days. B) Relative effectiveness between the highest and lowest value pixels with respect to three characteristics (as predicted by a multivariate model fit to the long-term drug effectiveness results).

delay, but also showed substantial increases in effectiveness with larger intervention radii (Fig 7B).

Using an intervention with lower efficacy or lower coverage (i.e., lower effective coverage) predictably has a proportionally lower overall efficacy (Fig 7C and 7D). However, we predict that medium to high final effectiveness can still be achieved with low efficacy interventions when used in a CATI strategy with larger case radii (250-1000m, Fig 7C and 7D). This is because larger rings ensure hotspots of transmission are treated regularly and repeatedly, increasing the chance of effective control. This prediction does, however, assume that interventions fail at random as opposed to more fundamental barriers to effectiveness, e.g., insecticide resistance or hard-to-reach human or mosquito populations.

CATI strategies are likely to be less effective in higher endemicity settings

Finally, we aimed to assess the generalisability of our findings to other dengue endemic settings outside Singapore where baseline transmission intensity is typically higher and case detection rates are lower (Fig 8). We predict that even modest increases in baseline transmission intensity (β_{μ}) considerably reduce the ability of even the largest ringed CATI strategies (1000m) to control dengue outbreaks (doubling β_{μ} leads to ~ 2 times the number of infections). Reductions in case detection probability also significantly restrict the utility of CATI strategies for outbreak response. Reducing detection probability by 50% (relative to Singapore levels) leads to 10–69% more cases with the biggest increases in low endemicity settings such as Singapore.

Discussion

CATI strategies are becoming increasingly popular for dengue control due to: i) a growing emphasis on outbreak prevention as the key goal of dengue control [41], ii) the failure of traditional control approaches to contain transmission long-term [24], iii) resource limitations and iv) increasing availability of data and data analytics platforms to predict and enable rapid precision responses [42]. There is, therefore, a growing need for a quantitative understanding of the strengths and limitations of CATI strategies to enable a more nuanced discussion of what role it can play in dengue control.

In this analysis we use a mathematical model, fit to a Singapore setting, to explore the impact of different CATI strategies. We find that controlling individual chains of transmission, e.g., within a household or local community, is likely to be highly time sensitive and prophylactic drugs could be a useful tool for extending this time window, particularly if people present



Fig 7. Scale, speed (A and B) and intervention effective coverage (C and D) required for a case-area targeted strategies with drugs (A and C) and vector control (B and D).

several days after the onset of symptoms. Following Phase 1 and 2 trials, more detailed modelling studies will be necessary to explore how the effectiveness of CATI strategies with prophylactic drugs changes with the specific characteristics (e.g., duration of efficacy) of each drug. Our results also have wider implications for index-case based intervention trial designs [43],





emphasising the need for rapid response to maintain intervention effect size, particularly for vector control tools that target adult mosquitoes [44]. We found that most areas treated one or more times using a CATI approach do see protection from dengue that lasts for the duration of the outbreak, especially in low density areas. Effectiveness of CATI strategies were found to be highly sensitive to the radius around the index case with radii 250m or larger giving the largest and most durable protection even if the efficacy of the interventions used is low. Finally, we estimate that CATI strategies will be highly sensitive to baseline transmission intensity and are only likely to be effective in areas that have already reduced transmission to lower levels

and have relatively high levels of case detection. This leads us to conclude that CATI strategies for dengue are likely to be effective, but are unlikely to be a precision, low resource requirement solution that is universally applicable. The need to have already achieved low levels of baseline transmission intensity, have high rates of case detection and be able to repeatedly deploy interventions at broad spatial scales suggest that CATI strategies will be best suited to non-endemic areas or as a "final-mile" approach in the latter stages of a wider dengue elimination programme. Similar conclusions have been reached on the role of reactive case detection and focal mass drug administration in malaria elimination [45, 46]—a vector-borne disease that is complicated by similar issues of asymptomatic infection and mobility of infected individuals. This also highlights the need for targeting preventative interventions such as environmental clean-up campaigns and vaccines (e.g., CDY-TDV) in high density, high transmission areas which may then allow CATI to contain outbreaks where it previously could not.

The predicted between-cluster variability in effectiveness of CATI also suggests that such strategies may need to be flexible and adaptable, changing index case radius, timeliness and even mixtures of interventions in response to the unfolding dengue epidemic. CATI strategies would, therefore, be best implemented alongside additional serological human or entomological sampling to further understand where and why effectiveness changes.

Our findings are subject to a number of limitations of our chosen model structure, fitting procedure and generalisability. Our model likely underrepresents geographic heterogeneity in mosquito-human contact, particularly in Singapore where small-scale construction sites are known to be significant contributors to overall transmission [47]. Higher geographic heterogeneity would make individual hotspots harder to control, but potentially more susceptible to focal targeting if they can be successfully identified. Over the time period of our data entomological surveys were only available from a small number of sentinel sites that were sufficient to measure variation in mosquito abundance over time, but not space at a sufficiently high resolution. If entomological data were available at higher spatial resolution the model may have been able to better capture the observed spatial heterogeneity in dengue cases [48]. Our model also demonstrated variable ability to fit different case clusters in Singapore, particularly outbreaks with more geographically restricted ranges. More work is needed to understand if and why the spatial spread of some epidemics differs. This may require developing a multi-serotype model to reflect co-circulation of multiple DENV serotypes and the complex pattern of human immunity that they impose in settings like Singapore. Integrating human movement data from, for example call data records [49], instead of relying on human moment models may also help better characterise movement heterogeneity across Singapore. Developing multipleinfection models with realistic population dynamics will also be important for projecting the longer-term reductions in disease burden attributable to CATI. This is important as it remains unknown whether the infections CATI averts are truly averted or just delayed until later in an individual's lifetime.

We also did not consider spatio-temporal variability in vector control when fitting our model. Our model placed no restrictions on the frequency within which interventions could be re-applied. Continual dosing of dengue prophylactic drugs over multiple months may not be feasible or practical. The model also assumed random efficacy and coverage of the intervention meaning multiple rounds of drug dosing or vector control boosted effective coverage and led us to the conclusion that large radii CATI strategies would be particularly effective. In reality there may be a cap on intervention effective coverage due to hard-to-reach populations/vector habitats or individuals/individual houses ineligible for drug dosing.

CATI strategies are becoming increasingly prevalent components of wider dengue control strategies. Here we aimed to quantitatively assess their potential effectiveness and assess their sensitivity to a range of operational and epidemiological parameters. This emphasised the

importance of using broad radii around index cases and the need to continue other interventions to keep baseline transmission intensity low and case detection rates high. If this can be achieved, CATI strategies with drugs or vector control could play an important role dengue control efforts.

Supporting information

S1 Fig. Contains a plot of the entomological and epidemiological data over time. (DOCX)

S2 Fig. Fit of the model to the data on human infectiousness over time. (DOCX)

S3 Fig. Map of estimated seroprevalence at baseline in Singapore. (DOCX)

S4 Fig. Model fit to each case cluster over successive rounds of Sequential Monte Carlo sampling.

(DOCX)

S1 Text. Describes how the population base maps were generated. (DOCX)

S2 Text. Equations for human movement models. (DOCX)

S1 Table. Sensitivity of clustering algorithm to choice of linkage break threshold. (DOCX)

Author Contributions

Conceptualization: Oliver J. Brady, Marnix Van Loock, Guillermo Herrera-Taracena, Joris Menten, W. John Edmunds, Stéphane Hué, Martin L. Hibberd.

Data curation: Yalda Jafari, Shuzhen Sim, Lee-Ching Ng.

Formal analysis: Oliver J. Brady, Yalda Jafari.

Funding acquisition: Stéphane Hué, Martin L. Hibberd.

Investigation: Oliver J. Brady, Marnix Van Loock, Guillermo Herrera-Taracena, Joris Menten, W. John Edmunds, Shuzhen Sim, Lee-Ching Ng, Stéphane Hué, Martin L. Hibberd.

Methodology: Oliver J. Brady, Adam J. Kucharski, Sebastian Funk, Yalda Jafari, W. John Edmunds, Shuzhen Sim, Lee-Ching Ng, Stéphane Hué, Martin L. Hibberd.

Supervision: Stéphane Hué, Martin L. Hibberd.

Validation: Oliver J. Brady, Adam J. Kucharski, Sebastian Funk, Yalda Jafari, Shuzhen Sim, Lee-Ching Ng.

Visualization: Oliver J. Brady.

Writing - original draft: Oliver J. Brady, Stéphane Hué, Martin L. Hibberd.

Writing – review & editing: Oliver J. Brady, Adam J. Kucharski, Sebastian Funk, Yalda Jafari, Marnix Van Loock, Guillermo Herrera-Taracena, Joris Menten, W. John Edmunds, Shuzhen Sim, Lee-Ching Ng, Stéphane Hué, Martin L. Hibberd.

References

- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, Moyes CL, et al. The global distribution and burden of dengue. Nature. 2013; 496: 504–507. https://doi.org/10.1038/nature12060 PMID: 23563266
- Stanaway JD, Shepard DS, Undurraga EA, Halasa YA, Coffeng LE, Brady OJ, et al. The global burden of dengue: an analysis from the Global Burden of Disease Study 2013. Lancet Infect Dis. 2016; 3099: 1–12. https://doi.org/10.1016/S1473-3099(16)00026-8 PMID: 26874619
- 3. World Health Organization. Dengue guidelines for diagnosis, treatment, prevention and control: new edition. 2009.
- Sim S, Ng LC, Lindsay SW, Wilson AL. A greener vision for vector control: The example of the Singapore dengue control programme. Kittayapong P, editor. PLoS Negl Trop Dis. 2020; 14: e0008428. https://doi.org/10.1371/journal.pntd.0008428 PMID: 32853197
- Anders KL, Nga LH, Van Thuy NT, Van Ngoc T, Tam CT, Tai LTH, et al. Households as Foci for Dengue Transmission in Highly Urban Vietnam. PLoS Negl Trop Dis. 2015; 9: e0003528. <u>https://doi.org/10.1371/journal.pntd.0003528 PMID: 25680106</u>
- Bowman LR, Donegan S, McCall PJ. Is dengue vector control deficient in effectiveness or evidence?: systematic review and meta-analysis. PLoS Negl Trop Dis. 2016; 10: e0004551. <u>https://doi.org/10.1371/journal.pntd.0004551</u> PMID: 26986468
- Liew C, Curtis CF. Horizontal and vertical dispersal of dengue vector mosquitoes, Aedes aegypti and Aedes albopictus, in Singapore. Med Vet Entomol. 2004; 18: 351–60. <u>https://doi.org/10.1111/j.0269-283X.2004.00517.x PMID</u>: 15642001
- Stoddard ST, Forshey BM, Morrison AC, Paz-Soldan VA, Vazquez-Prokopec GM, Astete H, et al. House-to-house human movement drives dengue virus transmission. Proc Natl Acad Sci U S A. 2013; 110: 994–9. https://doi.org/10.1073/pnas.1213349110 PMID: 23277539
- Perkins TA, Reiner RC, España G, Ten Bosch QA, Verma A, Liebman KA, et al. An agent-based model of dengue virus transmission shows how uncertainty about breakthrough infections influences vaccination impact projections. PLoS Comput Biol. 2019; 15: e1006710. https://doi.org/10.1371/journal.pcbi. 1006710 PMID: 30893294
- Low JGH, Ooi EE, Vasudevan SG. Current Status of Dengue Therapeutics Research and Development. J Infect Dis. 2017; 215: S96–S102. https://doi.org/10.1093/infdis/jiw423 PMID: 28403438
- Undurraga EA, Halasa YA, Shepard DS. Use of Expansion Factors to Estimate the Burden of Dengue in Southeast Asia: A Systematic Analysis. PLoS Negl Trop Dis. 2013; 7: e2056. <u>https://doi.org/10.1371/journal.pntd.0002056</u> PMID: 23437407
- Nguyen MN, Duong THK, Trung VT, Nguyen THQ, Tran CNB, Long VT, et al. Host and viral features of human dengue cases shape the population of infected and infectious Aedes aegypti mosquitoes. Proc Natl Acad Sci U S A. 2013; 110: 9072–7. https://doi.org/10.1073/pnas.1303395110 PMID: 23674683
- Liebman KA, Stoddard ST, Morrison AC, Rocha C, Minnick S, Sihuincha M, et al. Spatial Dimensions of Dengue Virus Transmission across Interepidemic and Epidemic Periods in Iquitos, Peru (1999–2003). PLoS Negl Trop Dis. 2012; 6: e1472. https://doi.org/10.1371/journal.pntd.0001472 PMID: 22363822
- Salje H, Lessler J, Maljkovic Berry I, Melendrez MC, Endy T, Kalayanarooj S, et al. Dengue diversity across spatial and temporal scales: Local structure and the effect of host population size. Science. 2017; 355: 1302–1306. https://doi.org/10.1126/science.aaj9384 PMID: 28336667
- Guzzetta G, Marques-Toledo CA, Rosà R, Teixeira M, Merler S. Quantifying the spatial spread of dengue in a non-endemic Brazilian metropolis via transmission chain reconstruction. Nat Commun. 2018; 9: 2837. https://doi.org/10.1038/s41467-018-05230-4 PMID: 30026544
- Barmak DH, Dorso CO, Otero M, Solari HG. Modelling interventions during a dengue outbreak. Epidemiol Infect. 2014; 142: 545–561. https://doi.org/10.1017/S0950268813001301 PMID: 23800514
- Nevai AL, Soewono E. A model for the spatial transmission of dengue with daily movement between villages and a city. Math Med Biol. 2014; 31: 150–178. https://doi.org/10.1093/imammb/dqt002 PMID: 23475426
- Ndeffo-Mbah ML, Durham DP, Skrip LA, Nsoesie EO, Brownstein JS, Fish D, et al. Evaluating the effectiveness of localized control strategies to curtail chikungunya. Sci Rep. 2016; 6: 23997. <u>https://doi.org/10.1038/srep23997 PMID: 27045523</u>
- Josseran L, Paquet C, Zehgnoun A, Caillere N, Le Tertre A, Solet J-L, et al. Chikungunya disease outbreak, Reunion Island. Emerg Infect Dis. 2006; 12: 1994–5. <u>https://doi.org/10.3201/eid1212.060710</u> PMID: 17354339
- ten Bosch QA, Clapham HE, Lambrechts L, Duong V, Buchy P, Althouse BM, et al. Contributions from the silent majority dominate dengue virus transmission. PLOS Pathog. 2018; 14: e1006965. https://doi. org/10.1371/journal.ppat.1006965 PMID: 29723307

- Stone CM, Schwab SR, Fonseca DM, Fefferman NH. Contrasting the value of targeted versus areawide mosquito control scenarios to limit arbovirus transmission with human mobility patterns based on different tropical urban population centers. PLoS Negl Trop Dis. 2019; 13: e0007479. <u>https://doi.org/10. 1371/journal.pntd.0007479</u> PMID: 31269020
- Ooi E-E, Goh K-T, Gubler DJ. Dengue prevention and 35 years of vector control in Singapore. Emerg Infect Dis. 2006; 12: 887–93. https://doi.org/10.3201/10.3201/eid1206.051210 PMID: 16707042
- Wong W-Y, Tan L-K, Ng L-C, Lam S, Low S-L, Teo D. Dengue seroprevalence of healthy adults in Singapore: serosurvey among blood donors, 2009. Am J Trop Med Hyg. 2015; 93: 40–45. https://doi.org/ 10.4269/ajtmh.14-0671 PMID: 26013376
- Hapuarachchi HC, Koo C, Rajarethinam J, Chong C-S, Lin C, Yap G, et al. Epidemic resurgence of dengue fever in Singapore in 2013–2014: A virological and entomological perspective. BMC Infect Dis. 2016; 16: 300. https://doi.org/10.1186/s12879-016-1606-z PMID: 27316694
- 25. Singapore National Environment Agency. Location of active clusters. [cited 12 Jan 2015]. Available: http://www.dengue.gov.sg/subject.asp?id=74
- 26. Ong J, Chong C-S, Yap G, Lee C, Abdul Razak MA, Chiang S, et al. Gravitrap deployment for adult Aedes aegypti surveillance and its impact on dengue cases. Aldridge RL, editor. PLoS Negl Trop Dis. 2020; 14: e0008528. https://doi.org/10.1371/journal.pntd.0008528 PMID: 32764763
- 27. Department of Statistics Singapore. Population Trends, 2020. Singapore; 2020. 2591-8028
- Simini F, González MC, Maritan A, Barabási A-L. A universal model for mobility and migration patterns. Nature. 2012; 484: 96–100. https://doi.org/10.1038/nature10856 PMID: 22367540
- Barbosa-Filho H, Barthelemy M, Ghoshal G, James CR, Lenormand M, Louail T, et al. Human Mobility: Models and Applications. Phys Rep. 2017; 734: 1–74. Available: http://arxiv.org/abs/1710.00004
- Amin FM, Kheong CC, Chem Y, Mudin RNB, Ng L-C, Lee K-S, et al. 2013 Dengue Outbreaks in Singapore and Malaysia Caused by Different Viral Strains. Am J Trop Med Hyg. 2015; 92: 1150–1155. https://doi.org/10.4269/ajtmh.14-0588 PMID: 25846296
- Lee KS, Lai YL, Lo S, Barkham T, Aw P, Ooi PL, et al. Dengue virus surveillance for early warning, Singapore. Emerg Infect Dis. 2010; 16: 847–9. https://doi.org/10.3201/eid1605.091006 PMID: 20409381
- Chan M, Johansson MA. The incubation periods of dengue viruses. PLoS One. 2012; 7: e50972. https://doi.org/10.1371/journal.pone.0050972 PMID: 23226436
- Manore CA, Hickmann KS, Xu S, Wearing HJ, Hyman JM. Comparing dengue and chikungunya emergence and endemic transmission in *A. aegypti* and *A. albopictus*. J Theor Biol. 2014; 356: 174–191. https://doi.org/10.1016/j.jtbi.2014.04.033 PMID: 24801860
- Brady OJ, Johansson MA, Guerra CA, Bhatt S, Golding N, Pigott DM, et al. Modelling adult Aedes aegypti and Aedes albopictus survival at different temperatures in laboratory and field settings. Parasites and Vectors. 2013; 6: 351. https://doi.org/10.1186/1756-3305-6-351 PMID: 24330720
- Duong V, Lambrechts L, Paul RE, Ly S, Lay RS, Long KC, et al. Asymptomatic humans transmit dengue virus to mosquitoes. Proc Natl Acad Sci U S A. 2015; 112: 14688–93. <u>https://doi.org/10.1073/pnas.</u> 1508114112 PMID: 26553981
- Johansson MA, Hombach J, Cummings DAT. Models of the impact of dengue vaccines: A review of current research and potential approaches. Vaccine. 2011; 29: 5860–5868. <u>https://doi.org/10.1016/j.vaccine.2011.06.042</u> PMID: 21699949
- Tan LK, Low SL, Sun H, Shi Y, Liu L, Lam S, et al. Force of Infection and True Infection Rate of Dengue in Singapore: Implications for Dengue Control and Management. Am J Epidemiol. 2019; 188: 1529– 1538. https://doi.org/10.1093/aje/kwz110 PMID: 31062837
- Schreiber MJ, Holmes EC, Ong SH, Soh HSH, Liu W, Tanner L, et al. Genomic epidemiology of a dengue virus epidemic in urban Singapore. J Virol. 2009; 83: 4163–73. https://doi.org/10.1128/JVI.02445-08 PMID: 19211734
- Siraj AS, Oidtman RJ, Huber JH, Kraemer MUG, Brady OJ, Johansson MA, et al. Temperature modulates dengue virus epidemic growth rates through its effects on reproduction numbers and generation intervals. PLoS Negl Trop Dis. 2017;11. https://doi.org/10.1371/journal.pntd.0005797 PMID: 28723920
- 40. Toni T, Welch D, Strelkowa N, Ipsen A, Stumpf MPH. Approximate Bayesian computation scheme for parameter inference and model selection in dynamical systems. J R Soc Interface. 2009; 6: 187–202. https://doi.org/10.1098/rsif.2008.0172 PMID: 19205079
- **41.** World Health Organization. Global strategy for dengue prevention and control, 2012–2020. World Health Organization; 2012.
- Runge-Ranzinger S, McCall PJ, Kroeger A, Horstick O. Dengue disease surveillance: an updated systematic literature review. Trop Med Int Heal. 2014; 19: 1116–1160. <u>https://doi.org/10.1111/tmi.12333</u> PMID: 24889501

- 43. Anders KL, Indriani C, Ahmad RA, Tantowijoyo W, Arguni E, Andari B, et al. The AWED trial (Applying Wolbachia to Eliminate Dengue) to assess the efficacy of Wolbachia-infected mosquito deployments to reduce dengue incidence in Yogyakarta, Indonesia: study protocol for a cluster randomised controlled trial. Trials. 2018; 19: 302. https://doi.org/10.1186/s13063-018-2670-z PMID: 29855331
- Reiner RC, Achee N, Barrera R, Burkot TR, Chadee DD, Devine GJ, et al. Quantifying the Epidemiological Impact of Vector Control on Dengue. PLoS Negl Trop Dis. 2016; 10: e0004588. <u>https://doi.org/10. 1371/journal.pntd.0004588</u> PMID: 27227829
- 45. Gerardin J, Bever CA, Bridenbecker D, Hamainza B, Silumbe K, Miller JM, et al. Effectiveness of reactive case detection for malaria elimination in three archetypical transmission settings: a modelling study. Malar J. 2017; 16: 248. https://doi.org/10.1186/s12936-017-1903-z PMID: 28606143
- 46. Brady OJ, Slater HC, Pemberton-Ross P, Wenger E, Maude RJ, Ghani AC, et al. Role of mass drug administration in elimination of *Plasmodium falciparum* malaria: a consensus modelling study. Lancet Glob Heal. 2017; 5: e680–687. https://doi.org/10.1016/S2214-109X(17)30220-6 PMID: 28566213
- Liang S, Hapuarachchi HC, Rajarethinam J, Koo C, Tang C-S, Chong C-S, et al. Construction sites as an important driver of dengue transmission: implications for disease control. BMC Infect Dis. 2018; 18: 382. https://doi.org/10.1186/s12879-018-3311-6 PMID: 30089479
- Ong J, Liu X, Rajarethinam J, Kok SY, Liang S, Tang CS, et al. Mapping dengue risk in Singapore using Random Forest. PLoS Negl Trop Dis. 2018; 12: e0006587. <u>https://doi.org/10.1371/journal.pntd.</u> 0006587 PMID: 29912940
- 49. Wesolowski A, Qureshi T, Boni MF, Sundsøy PR, Johansson MA, Rasheed SB, et al. Impact of human mobility on the emergence of dengue epidemics in Pakistan. Proc Natl Acad Sci U S A. 2015; 112: 11887–11892. https://doi.org/10.1073/pnas.1504964112 PMID: 26351662